Author’s response to reviews

Title: Retinal and Choroidal Thickness in Paediatric Patients with Hypoalbuminaemia Caused by Nephrotic Syndrome

Authors:

Wenbo Zhang (80777785@qq.com)
Yadi Zhang (zhangyadi0820@163.com)
Lei Kang (kanglei105@163.com)
Xiaopeng Gu (guxp1988@foxmail.com)
Hailong Wu (whl0210@126.com)
Liu Yang (lucy02114@163.com)

Version: 2 Date: 16 Jan 2019

Author’s response to reviews:

Dear editor(s),

Thank you very much for your letter and advice. We have revised the paper, and would like to re-submit it for your consideration. We hope that the revision is acceptable and look forward to hearing from you soon.

With best wishes,

Yours sincerely,

Liu Yang

Replies are as follows:

1. We are very sorry for the overlaps. We have re-phrased these parts very carefully and the article published by Yan was cited.

(The modified section are as follows:

Abstract section, line 2, 4, 9-13, 16-18, page 2; line 3, page 3;
2. Written informed consent was obtained from all participants. And this was clearly stated in the manuscript. (Methods section, line 17-18, page 5)

3. The funding body had no influence on the design, collection, analysis or interpretation of the data or on the writing of the manuscript. And this was clearly stated in the manuscript. (Funding section, line 6-8, page 18)

4. We have provided more justification of the contributions for YDZ, LK and XPG. YDZ were responsible for the project design, while LK and XPG for the data analysis. This was clearly stated in the manuscript. (Authors’ contributions section, line 10-15, page 18)

5. We used the initials of the authors’ names and removed the title. (Authors’ contributions section, line 10-15, page 18)

6. The current submission is a clean version without any change tracks, highlights, or text in different colors.